Atmospherically Driven Seasonal and Interannual Variability in the Lagrangian Transport Time Scales of a Multiple-inlet Coastal System

Jeancarlo Manuel Fajardo-Urbina¹, Gijs Arts¹, Ulf Gräwe², Herman J.H. Clercx¹, Theo Gerkema³, and Matias Duran-Matute¹

¹Eindhoven University of Technology

 2 Leibniz Universität Hannover, Leibniz-Institut für Ostseeforschung Warnemünde
 $^3{\rm NIOZ}$ Royal Netherlands Institute for Sea Research

December 8, 2022

Abstract

Intense short-term wind events can flush multiple-inlet systems and even renew the water entirely. Nonetheless, little is known about the effect of wind variations at seasonal and interannual scales on the flushing of such systems. Here, we computed two Lagrangian transport time scales (LTTS), the residence and exposure times, for a multiple-inlet system (the Dutch Wadden Sea) over 36 years using a realistic numerical model simulation. Our results reveal pronounced seasonal and interannual variability in both system-wide LTTS. The seasonality of the LTTS is strongly anti-correlated to the wind energy from the prevailing directions, which are from the southwesterly quadrant and coincidentally aligned with the geographical orientation of the system. This wind energy, which is stronger in autumn-winter than in spring-summer, triggers strong flushing (and hence low values of the LTTS) during autumn-winter. The North Atlantic Oscillation (NAO) and the Scandinavia Pattern (SCAN) are shown to be the main drivers of interannual variability in the local wind and, ultimately, in both LTTS. However, this coupling is much more efficient during autumn-winter when these patterns show larger values and variations. During these seasons, a positive NAO and a negative SCAN induce stronger winds in the prevailing directions, enhancing the flushing efficiency of the system. The opposite happens during positive SCAN and negative NAO, when weaker flushing during autumn-winter is observed. Thus, large-scale atmospheric patterns strongly affect the interannual variability in flushing and are potential drivers of the long-term ecology and functioning of multiple-inlet systems.

Atmospherically Driven Seasonal and Interannual Variability in the Lagrangian Transport Time Scales of a Multiple-inlet Coastal System

Jeancarlo M. Fajardo-Urbina¹, Gijs Arts¹, Ulf Gräwe², Herman J.H. Clercx¹, Theo Gerkema³, Matias Duran-Matute¹

¹Fluids and Flows group and J.M. Burgers Center for Fluid Dynamics, Department of Applied Physics and Science Education, Eindhoven University of Technology, Eindhoven, the Netherlands ²Leibniz Institute for Baltic Sea Research Warnemünde, Rostock, Germany ³Department of Estuarine and Delta Systems, NIOZ Royal Netherlands Institute for Sea Research, Yerseke, the Netherlands

Key Points:

1

2

3

5

6

7 8

9

10

11

17

12	•	The Lagrangian transport time scales in the Dutch Wadden Sea are typically 1.8
13		times smaller in autumn-winter than in to spring-summer.
14	•	The seasonal and interannual variability of the Lagrangian transport time scales
15		is attributed to the local wind.
16	•	The winter interannual variations are well explained by North Atlantic large-scale

atmospheric patterns.

Corresponding author: Matias Duran-Matute, m.duran.matute@tue.nl

18 Abstract

Intense short-term wind events can flush multiple-inlet systems and even renew the wa-19 ter entirely. Nonetheless, little is known about the effect of wind variations at seasonal 20 and interannual scales on the flushing of such systems. Here, we computed two Lagrangian 21 transport time scales (LTTS), the residence and exposure times, for a multiple-inlet sys-22 tem (the Dutch Wadden Sea) over 36 years using a realistic numerical model simulation. 23 Our results reveal pronounced seasonal and interannual variability in both system-wide 24 LTTS. The seasonality of the LTTS is strongly anti-correlated to the wind energy from 25 the prevailing directions, which are from the southwesterly quadrant and coincidentally 26 aligned with the geographical orientation of the system. This wind energy, which is stronger 27 in autumn-winter than in spring-summer, triggers strong flushing (and hence low val-28 ues of the LTTS) during autumn-winter. The North Atlantic Oscillation (NAO) and the 29 Scandinavia Pattern (SCAN) are shown to be the main drivers of interannual variabil-30 ity in the local wind and, ultimately, in both LTTS. However, this coupling is much more 31 efficient during autumn-winter when these patterns show larger values and variations. 32 During these seasons, a positive NAO and a negative SCAN induce stronger winds in 33 the prevailing directions, enhancing the flushing efficiency of the system. The opposite 34 happens during positive SCAN and negative NAO, when weaker flushing during autumn-35 winter is observed. Thus, large-scale atmospheric patterns strongly affect the interan-36 nual variability in flushing and are potential drivers of the long-term ecology and func-37 tioning of multiple-inlet systems. 38

³⁹ Plain language summary

In multiple-inlet coastal systems, strong wind events efficiently renew the water in 40 these systems. In this paper, we investigate if the flushing of such systems has also a marked 41 response to wind variability at longer time scales. To quantify the flushing, we compute 42 the time that particles, each representing a certain volume of water, spend in the sys-43 tem before leaving it (known as the residence time) and the total time they spend within 44 it considering future returns (known as the exposure time). Our 36-year simulation of 45 the hydrodynamics of the DWS shows that the wind induces seasonal and interannual 46 variations in both spatially-averaged quantities. The seasonality is related to the wind 47 energy from the dominant directions, which is much larger during autumn-winter than 48 during spring-summer. This variation leads to a reduction of both time scales by, on av-49 erage, a factor 1.8 from spring-summer to autumn-winter. Two well-known North At-50 lantic large-scale atmospheric patterns, primarily active during autumn-winter, induce 51 interannual variations in the wind and consequently in both time scales. Thus, future 52 changes in these patterns could strongly affect water transport and the ecology of the 53 Dutch Wadden Sea. Similar situations are likely to occur in other multiple-inlet systems. 54

55 1 Introduction

Transport time scales (TTS), such as the residence, exposure, transit, age, and flush-56 ing times (Zimmerman, 1976; Monsen et al., 2002), are measures for the efficiency of trans-57 port and exchange of water or freshwater content within a water body system and with 58 its surroundings (Cucco et al., 2009; Duran-Matute et al., 2014; Rayson et al., 2016; Xiong 59 et al., 2021). They also serve to estimate the time that a substance, like dissolved nitro-60 gen, takes to be transported off-shore from high-productivity coastal regions (Hailegeorgis 61 et al., 2021); to understand the variability of the mineralization rates of organic matter 62 in sediments (den Heyer & Kalff, 1998); to explain regional differences of nutrient and 63 eutrophication levels (González et al., 2008; Schwichtenberg et al., 2017); and as a first-64 order estimation of the exposure of a region (e.g. a protected area) to pollutants (Soomere 65 et al., 2011; Patgaonkar et al., 2012; Pawlowicz et al., 2019). 66

Depending on a coastal system's particularities, the TTS's variability can be highly 67 affected by tides, freshwater discharge, gravitational circulation, winds, and other fac-68 tors. The influence of some of these forcing mechanisms on the intra-annual and the sea-69 sonal variability of the TTS has been explored in bights (Zhang et al., 2010), bays (Dippner 70 et al., 2019; Jiang et al., 2019) and lakes (Cimatoribus et al., 2019). However, these stud-71 ies were based on just 1 to 2 years of data, and thus, a robust relationship of the sea-72 sonality with the local forcing cannot be expected if there is a marked interannual vari-73 ability. 74

75 A realistic simulation covering 32 years was used by Du and Shen (2016) to study the residence time in the Chesapeake Bay. The seasonal, monthly and interannual vari-76 abilities of the system-wide Eulerian residence time were found to be mainly controlled 77 by the freshwater discharge. To determine the role of the wind, they compared two sim-78 ulations for a given year, one with the full forcing and the other without wind. They found 79 that downstream and upstream winds reduce the residence time in the eastern side of 80 the Bay, whereas only upstream winds increase the residence time on the opposite side. 81 This means that in this single-inlet system winds from different directions can trigger 82 complex patterns in the TTS but not necessarily induce net transport across the system. 83

Single-inlet systems contrast with multiple-inlet systems because, in the latter, winds 84 from specific directions are very efficient in forcing net residual transport across the sys-85 tem (Li, 2013; Herrling & Winter, 2015; Duran-Matute et al., 2016). Due to this effect, 86 the influence of other forcing mechanisms can become of secondary importance during 87 strong wind conditions. Thus, winds in multiple-inlet systems can strongly modify the 88 TTS at local, inter-basin, and system-wide scales. This effect has been observed in dif-89 ferent multiple-inlet systems using numerical simulations. Cucco and Umgiesser (2006) 90 showed that, in the Venice lagoon, strong northeasterly bora winds (of around 12 m/s) 91 lead to a fully wind-driven dominated system, to a reduction of the system-averaged res-92 idence time by a factor of 3, and to a negligible return flow. In the Dutch Wadden Sea 03 (DWS), strong winds exceeding 10 m/s, and aligned with the geographical orientation of the system, induce a wind-driven flow that reduces the system-wide flushing time of 95 freshwater discharge by a factor of 10-15 (Duran-Matute et al., 2014; Donatelli et al., 96 2022a). Similar strong winds as in the previous cases, also reduced the monthly-average 97 residence time by about a factor 2 in the Virginia Coast Reserve (Safak et al., 2015); and 98 the daily-average Lagrangian residence time (for particles released every 1h during a par-99 ticular day) in areas located between the inlets of the Barnegat Bay-Little Egg Harbor 100 estuary by a factor between 2-4 (Defne & Ganju, 2015). 101

Until now, the previous studies linking TTS to wind in multiple-inlet systems fo-102 cused on idealized fixed wind conditions (e.g. Cucco & Umgiesser, 2006), synoptic-scale 103 events (e.g Duran-Matute et al., 2014; Safak et al., 2015) and annual statistics (e.g Do-104 natelli et al., 2022a). In the latter case, Donatelli et al. (2022a) showed that sporadic strong 105 high-frequency winds (with time scales in the order of days) could impact the annual TTS 106 averages in the DWS, but also the long-term values (mean or median representative of 107 their 11-year simulation). However, they did not isolate the effect of high- and low-frequency 108 winds (with time scales of months or longer) on the TTS to unequivocally attribute the 109 changes in the annual and long-term TTS to high-frequency wind events. The relevance 110 of the low-frequency variability is further suggested by the fact that monthly and multi-111 decadal sea level variability in the North Sea region is modulated by large-scale atmo-112 spheric patterns, which are represented by the North Atlantic Oscillation (NAO), the 113 East Atlantic Pattern (EAP) and the Scandinavia Pattern (SCAN) (Chafik et al., 2017; 114 Frederikse & Gerkema, 2018). Therefore, we investigate if and how much these large-115 scale atmospheric patterns affect the TTS in the DWS. 116

Our goal is to determine the low-frequency variability (i.e., the seasonality and interannual variations) of the Lagrangian TTS (LTTS), particularly the residence and exposure times, in a multiple-inlet system. Moreover, we aim to correlate their system-wide



Figure 1. Map of the region of interest. The red contour surrounds most of the DWS and denotes the region where particles were deployed. The names of the five inlets are indicated in black. The Schiermonnikoog and the Terschelling watersheds are marked in green. The location and the names of the two main sluices are depicted in blue. The location of the stations employed for the validation with the sea-surface height (SSH) are shown in magenta. The color bar denotes the depth.

behavior with the wind and large-scale atmospheric patterns. The region of analysis cov-120 ers most of the DWS (Figure 1): a UNESCO world heritage site and a complex multiple-121 inlet system. Due to the lack and the difficulty of acquiring observed Lagrangian data 122 in shallow coastal regions, the results are based on a realistic 36-year simulation (1980-123 2015) of the DWS, combined with particle tracking. The simulation consists of an of-124 fline coupling of the General Estuarine Transport Model (GETM; Burchard & Bolding, 125 2002) with the Probably A Really Efficient Lagrangian Simulator (Parcels) v2.1.1 (Lange 126 & van Sebille, 2017; Delandmeter & Van Sebille, 2019). 127

¹²⁸ 2 Data and methods

129

2.1 Numerical models

130 2.1.1 Eulerian model

The currents, sea level, salinity, temperature, and density are obtained through three-131 dimensional, baroclinic numerical simulations performed using GETM. The setup is based 132 on four nested models, with the DWS numerical domain as the end-member. The do-133 main is discretized using an equidistant grid of 200 m resolution using the Rijksdriehoek 134 projection (the standard projection employed by the Dutch Government) in the horizon-135 tal and 25 layers in the vertical. The bathymetry was built based on the measurements 136 closest in time to 2009-2010 (see Duran-Matute et al., 2014, for details), and the result-137 ing map was kept fixed throughout the 36-year simulation. This was done intentionally 138 to remove the effects of bathymetry variations on the hydrodynamics of the system and 139 to focus on the role of the atmospheric forcing. The meteorological forcing was taken from 140

the dataset "Uncertainties in Ensembles of Regional Reanalyses" (UERRA; Ridal et al., 2017), which has a spatial resolution of 11 km and a temporal resolution of 1 h. The freshwater discharge through the Den Oever and Kornwerderzand sluices and 10 other smaller ones was reconstructed based on data from Rijkswaterstaat with a temporal resolution of 12 minutes (see Duran-Matute et al., 2014, for details). Our model configuration is almost identical to those employed by Donatelli et al. (2022a, 2022b), but the simulation here spans 36 years instead of 11 years.

We contrast our numerical results with sea-surface height (SSH) measured at 14 tidal stations located within and around the DWS (Figure 1). Our simulation performed similar as the one by Duran-Matute et al. (2014), and a full description of the validation can be found in Text S1 from Supporting Information S1.

2.1.2 Lagrangian model

152

Passive particle trajectories were obtained offline by feeding vertically-averaged ve-153 locities every 20 minutes from the GETM simulation to Parcels. We used a fourth-order 154 Runge-Kutta method for the temporal integration and a bilinear interpolation in space, 155 which showed to be accurate enough in idealized and realistic applications (Lange & van 156 Sebille, 2017; Delandmeter & Van Sebille, 2019). We used a time step of 158 s to bal-157 ance accuracy and computational time. It was also chosen to have the timestep as an 158 integer fraction of the M2 tidal period (44714 s), which is the main tidal constituent in 159 the DWS. In our setup, particles were released within the region of interest (denoted with 160 the red contour in Figure 1) in the center of each of the 200 m \times 200 m grid cells but 161 skipping every other cell. Then they were advected with depth-averaged currents to cap-162 ture the effects of the net horizontal currents on water transport. This procedure was 163 repeated every M2 period from January-1980 to October-2015, with each release con-164 sisting of 12967 particles. The total amount of particles trajectories obtained is (12967 165 particles per deployment) \times (25290 deployments) \approx 328 million particle trajectories (\approx 166 1.1 TB of data). To avoid deploying particles when most of the tidal flats are dry, the 167 first deployment was near the time of maximum water volume within the DWS so that 168 the subsequent ones (every M2 period) were also close to maximum volume conditions. 169 The particle positions were saved every M2 period to remove the back-and-forth due to 170 this dominant semidiurnal tidal constituent in the DWS (Zimmerman, 1976). We, thus, 171 capture the net residual displacement of the particles. We note that individual tidal pe-172 riods may deviate somewhat from the M2 period, but since M2 is the dominant constituent, 173 the long-term mean tidal period equals the M2 period (Gerkema, 2019). 174

To avoid errors in the estimation of LTTS due to particles being stuck because of 175 being released too close to the coast or to areas that seldom flood, we removed such par-176 ticles from our original dataset (containing ≈ 328 million particles trajectories) using three 177 steps. In the first step, we discarded beaching particles (defined as the ones located within 178 100 m of a land point at any time), which represents around 8.4% of the original data. 179 In the second step, we removed particles that do not leave our domain of interest (red 180 contour in Figure 1) through its open boundaries within their integration time (around 181 1.8%). This latter condition help to remove particles whose trajectories can be poten-182 tially affected by the poorly resolved flow near the coast, even though they are not beach-183 ing according to our definition. These particles can spend some days barely moving and 184 meandering close to the coast due to the small currents present in these areas. In gen-185 eral, these first two steps remove most of the particle trajectories that suffer from nu-186 merical artifacts (e.g. error of the numerical solvers, the spatial resolution of the flow, 187 and the temporal time step for the integration of trajectories; which are described by Delandmeter 188 and Van Sebille (2019)). In the third and last step, all the particles released from po-189 sitions in which the amount of discarded particles (from the previous two steps) repre-190 sents more than 30% of the total deployments per point of release were also discarded. 191 This step, removes an extra 3.4% of particles. These particles were mostly deployed in 192

the few regions that are above mean sea level, which are only flood during large storm surges. However, omitting this last step leads to almost the same results because most of the problematic particles were already removed using the first two steps. Finally, after applying all the previous steps, we end up with around 283 million particle trajectories for our analysis.

To check the sensitivity of our results when using non-uniform total integration times, 198 trajectories of particles released at the beginning of every month of our 36-year simu-199 lation were integrated for 177 M2 periods (about 91 days). Then, we decreased this time 200 linearly until 117 M2 periods (around 60 days) for the particles released at the end of 201 every month. Particles were not tracked anymore if they crossed the boundaries of the 202 numerical domain before their integration time was reached. We found that under a com-203 mon integration time of 60 days, instead of the 60-91 days interval employed in our anal-204 ysis, the results were almost the same since 98.5 % of the 283 million particles left our 205 domain of interest (see red contour in Figure 1) trough its open boundaries before 60 days. 206

207

2.2 Definitions of Lagrangian transport time scales (LTTS)

The Lagrangian residence time is a function of space and time and highlights the 208 spatio-temporal heterogeneity of transport. It is defined as the time required for a par-209 ticle to exit a domain for the first time (Zimmerman, 1976; Monsen et al., 2002). Nonethe-210 less, this first-crossing definition has a drawback. When particles are close to an open 211 boundary, they might exit the system during ebb and return during flood, possibly re-212 peating this behavior during the following cycles, after which they can remain in the do-213 main for several days. In this way, such a definition of the residence time might give a 214 wrong idea of the actual time particles spend in the system, particularly close to the in-215 lets. We largely avoid this problem by saving particle positions only every M2 period (i.e., 216 using the net or residual displacement). With those generated tracks, we define the La-217 grangian residence time as the number of M2 periods required for the particles to leave 218 our domain of interest (red contour in Figure 1) trough its open boundaries. Since the 219 residence time varies with space and time, we define T_r^{ij} as the residence time of a par-220 ticle released during the *j*-th deployment (at time t_j) at position (x_i, y_i) , where *i* is the 221 spatial index of the particle released in the center of the 200 m \times 200 m grid, and t_i are 222 the times of deployments (every M2 cycle during our 36-year simulation). A similar ap-223 proach is employed for the Lagrangian exposure time T_e^{ij} , which is defined as the total 224 amount of time a particle spends in our system (neglecting the time spent outside of it), 225 and thus $T_e^{ij} \ge T_r^{ij}$ (Monsen et al., 2002; Huguet et al., 2019). 226

To describe the spatial variability between seasons, we further define the temporal average over N_d^i deployments as

$$T_{r}^{i} = \frac{\sum_{j=1}^{N_{d}^{i}} T_{r}^{ij} H^{ij}}{\sum_{j=1}^{N_{d}^{i}} H^{ij}},$$
(1)

where N_d^i is the total amount of deployments per point of release available during time period for averaging, and H^{ij} is the height of the water column in which the particle is deployed. Specifically, we consider two temporal averages: one for all autumn-winter (September-February) and one for all spring-summer (March-August) seasons of our 36-year simulation. The weighted average using H^{ij} is employed because particles are advected with depth-averaged currents, and thus, a particle released over a large water column represents more fluid with that value of T_r^{ij} (Ridderinkhof & Zimmerman, 1990). To study the variability of the system-wide LTTS, we define the spatial average over all N_p^j particles released at the same time as

$$T_{r}^{j} = \frac{\sum_{i=1}^{N_{p}^{j}} T_{r}^{ij} H^{ij}}{\sum_{i=1}^{N_{p}^{j}} H^{ij}}.$$
(2)

Similarly, we obtain T_e^i and T_e^j using equivalents to equation (1) and equation (2) for the exposure time, respectively.

229 **2.3** Atmospheric forcing characterization

To understand the origin of the variability of the LTTS, we characterize the atmospheric forcing using a local and a large-scale approach.

2.3.1 Local approach

232

For the local approach, we employ the concept of sectorial wind energy, following Gerkema and Duran-Matute (2017). The wind direction is divided into eight sectors using the indices $s = 1, \ldots, 8$, corresponding to southerly (S), southeasterly (SE), east-erly (E), northeasterly (NE), northerly (N), northwesterly (NW), westerly (W), and south-westerly (SW) winds (i.e., the direction from which the wind blows). Then, the kinetic energy of an air parcel (wind energy) with mass m crossing a unit area A during an interval Δt and from sector or direction s is given by

$$E_{s,n} = \frac{1}{2}mW_{s,n}^2 = \frac{1}{2}\rho V W_{s,n}^2 = \frac{1}{2}\rho A\Delta t W_{s,n}^3,$$
(3)

where V is the volume, which is equal to the area A times the length $W_{s,n}\Delta t$; $W_{s,n}$ is the hourly wind speed (used in the GETM simulation) blowing from sector s, with n as a temporal index running over our full 36-year simulation; $\Delta t = 3600$ s is the resolution of our wind data; and $\rho = 1.225$ kg m⁻³ is the density of the air at sea level with temperature of 15°.

In all our analysis, the wind energy from the grid point closest to the middle of the Texel inlet is employed. Due to the small spatial variations of the wind inside the DWS, we anticipate that using the wind energy from different locations does not change qualitatively our results, as was also the case for Duran-Matute et al. (2016) in their analysis of the residual volume transport in the DWS.

243 2.3.2 Large-scale approach

For the large-scale approach, we use the North Atlantic Oscillation (NAO), the East 244 Atlantic Pattern (EAP), and the Scandinavian Pattern (SCAN). To derive them, we per-245 form an empirical orthogonal function (EOF) analysis following Chafik et al. (2017) and 246 Frederikse and Gerkema (2018). With this method, the atmospheric patterns have spa-247 tial structures represented by empirical orthogonal functions (EOFs), whereas their tem-248 poral variability are captured by principal components (PCs). To obtain the EOFs and 249 PCs, we employ the monthly-mean sea level pressure (SLP) from the NCEP/NCAR Re-250 analysis 1 (Kalnay et al., 1996) spanning the period 1950-2015 in the North Atlantic/European 251 sector (30°-80°N, 80W-50°E). For every grid cell, the monthly-mean SLP is detrended 252 and deseasonalized, i.e., the linear trend, and the annual and semi-annual components 253 are removed. Then, the data are weighted by the cosine of the latitude at every grid point 254 before computing the EOF analysis. This is done to give less weight to grid cells located 255 towards the poles as they represent less area (which decreases with the cosine of the lat-256 itude in spherical coordinates). Finally, the EOF analysis is performed only in the North 257 Atlantic domain to avoid the influence of regions outside of it in the three main modes 258 of variability obtained. 259

²⁶⁰ 3 Results and Discussions

261

3.1 Seasonality and interannual variability of the LTTS

The mean autumn-winter and spring-summer spatial patterns for the residence time 262 T_r^i (from equation (1)) are shown in Figures 2a and 2b, and in Figures 2c and 2d for the 263 exposure time T_e^i . The lowest values are found near the inlets since they are the primary 264 regions for exchange with the adjacent North Sea. Particles deployed around them leave 265 the system in less than one week, with the corresponding areas being larger during autumn-266 winter. The highest values are found farther into the basins and mostly in the Western 267 DWS (west of the Terschelling watershed). These values are up to a factor of two larger 268 during spring-summer than during autumn-winter. Most of the particles deployed in the 269 Western DWS during spring-summer tend to return to the DWS (see difference between 270 T_e^i and T_r^i in the inset of Figure 2d). Nonetheless, during autumn-winter (inset of Fig-271 ure 2c) this effect is observed only in the southern-most part of the domain. Consistent 272 with all of the previously mentioned behavior, the wind roses show a marked difference 273 between autumn-winter (Figure 2e) and spring-summer (Figure 2f), with the former ex-274 hibiting more frequent and stronger winds from the W, SW and S directions. 275

To get a representative seasonal cycle of the LTTS in the full DWS, we computed 276 the spatial mean of the residence and exposures times (i.e., T_r^j and T_e^j from equation (2)). 277 Then, the annual cycle for the residence and exposure times are estimated by fitting T_{x}^{2} 278 and T_e^j to a model with a free constant and an annual harmonic. The system-wide an-279 nual signal of T_r^j varies from 10-11 days in November-January to 17-18 days in May-July, 280 and for T_e^j from 14-16 days to 24-25 days, respectively (Figure 3). This means that the 281 extra time that particles spend in the DWS system after they leave for the first time is 282 smaller in November-January (around 4 days) than during May-July (around 7 days). 283

To understand the variability superimposed on the seasonal cycle, high-frequency 284 effects (e.g., tides and energetic synoptic-scale events) from T_r^j (which has an M2 res-285 olution) were removed by computing a 15-day mean, which is shown as T_r in Figure 4a. 286 Afterwards, we performed a wavelet analysis (Torrence & Compo, 1998) of this spatially-287 averaged 15-day-mean residence time (T_r) , using the rectification of the bias proposed 288 by Liu et al. (2007), to capture the localized time-frequency information in our time se-289 ries. The wavelet power spectrum exhibits the strongest signal around the annual pe-290 riod (Figure 4b). However, anomalous behavior is still observed, with periods display-291 ing a strong annual power (e.g., around 1983, 1990, 2000, and 2014) or a weak one (e.g., 292 around 1986, 1996, 2006, and 2010). Similar results are obtained for the equivalent ex-293 posure time T_e (Figure 5). Clearly, studies of the DWS based on time series of a few years, 294 like those for 2009-2011 by Duran-Matute et al. (2014, 2016) and for 2005-2015 by Donatelli 295 et al. (2022a, 2022b), cannot capture well this rich temporal variability of the system-296 wide transport characteristics. 297

The wavelet power spectrum of \hat{T}_r (Figure 4b) also contains significant power out-298 side the annual signal, like the time spans with strong four-month periodicity around 1984, 299 1990 and 1997. There are also higher frequency events with a still significant signal but 300 they are close to the background noise. These events cause large peaks with a relatively 301 low persistence of only a few weeks. Thus, to focus on the system-wide low-frequency 302 (seasonal and interannual) variations of the LTTS and to find links with the wind forc-303 ing (which we discuss in section 3.2) and large-scale circulation and atmospheric patterns 304 (which will be addressed in section 3.3), we filtered the time series. We removed vari-305 ability from T_r using a wavelet filter with a cutoff period of half a year. This procedure 306 resulted in the half-year low-pass filtered signal of the spatially-averaged 15-day-mean 307 residence time (T_r) and exposure time (T_e) (see Figures 4a and 5a, respectively). Most 308 of the variability at low frequencies is due to the seasonal cycle. However, there are fluc-309 tuations at interannual time scales that modulate it. Clear examples are the anomalous 310 winters (DJF) with the lowest T_r (5-7 days) of 1983, 1990, 1995, 2000, 2007, 2008, and 311



Figure 2. The time-averaged residence time T_r^i for a) autumn-winter (September-February) and b) spring-summer (March-August) based on the 36-year simulation; and the mean exposure time T_e^i for c) autumn-winter (September-February) and d) spring-summer (March-August). The insets in c) and d) show the difference between T_e^i and T_r^i . Regions in white within the DWS were removed from the analysis (see section 2.1.2). The grey line indicates the -5 m isobath. e) Autumn-winter and f) spring-summer wind rose, in which the purple numbers indicate the percentage of time that the wind blows from a particular direction.



Figure 3. Annual cycle of the residence time (red), the exposure time (magenta), and the wind energy (green).

³¹² 2014 (using the year after December as the name of the winter); or the anomalous winters with the largest \tilde{T}_r (12-17 days) of 1996, 2003, 2006, 2009, and 2010. In summer (JJA), the variability of the peaks is less pronounced, with values that vary between 15 and 20 days. For \tilde{T}_e (Figure 5a), a similar behavior is observed during those winters, with the lowest values around 7-8 days and the largest between 16-28 days. During summer, \tilde{T}_e mainly varies between 21 and 30 days.

3.2 Impact of the wind on the system-wide LTTS

318

To show the dominance of the wind on the variability of the LTTS, we propose a reconstruction of \tilde{T}_r (and an identical one for \tilde{T}_e) using the energy of the most dominant wind sectors (W, SW, and S). Winds from these directions are the most efficient for driving a strong residual flow from the Texel inlet to the Vlie inlet and the Terschelling watershed (Duran-Matute et al., 2014). We refer to this reconstruction as the *wind-based model* and is given by

$$\overline{T}_r = A e^{-E/B},\tag{4}$$

where \tilde{E} is the sum of the half-year low-pass filter signal of the 15-day-mean wind en-319 ergy of the dominant sectors (see Appendix A for the definition of the 15-day-mean wind 320 energy per sector, and for the computation of \tilde{E}). Because \tilde{T}_r and \tilde{T}_e are quantities that 321 depend on the future, the wind-based model employs \tilde{E} of the next 15-day interval in com-322 parison to the LTTS time series. The constants A and B are fitting coefficients. The neg-323 ative sign in the argument of the exponential reflects the anti-correlation between T_r and 324 \hat{E} (Figure 6a), which means that strong \hat{E} conditions result in low \hat{T}_r and \overline{T}_r ; while the 325 opposite holds during weak \tilde{E} conditions. The constant $A = 19.29 \pm 0.16$ days (with 326 95% CI) for \overline{T}_r represents the maximum value that can be predicted with \overline{T}_r , which is 327 reached during E = 0 conditions. This constant contains the mean effects of the resid-328 ual tides, freshwater discharge, and other wind directions not included in the reconstruc-329 tion. The constant $B = 2.31 \pm 0.06$ MJ is an e-folding wind energy scale for \overline{T}_r , which 330 indicates that an increase in \tilde{E} equal to B would lead to a reduction of \overline{T}_r by 63%. For 331 the exposure time, there is also a strong anti-correlation between \tilde{T}_e and \tilde{E} (Figure 6b). 332 The maximum value predicted by \overline{T}_e is given by $A = 27.96 \pm 0.13$ days, and its e-folding 333 wind energy scale is $B = 2.07 \pm 0.03$ MJ. 334



Figure 4. (a) Time series of the spatially-averaged 15-day-mean residence time (\hat{T}_r) , and its half-year low-pass filtered component (\tilde{T}_r) . (b) Wavelet power spectrum of \hat{T}_r , where the black contour encloses regions with power greater than a lag-1 red-noise process with 95% confidence level; and the grey shadow region is the "cone of influence", where errors due to the finite length of the time series are present. The horizontal red dashed line highlights the half-year period employed as a cutoff for computing \tilde{T}_r .

The values of \overline{T}_r (Figure 6a) match the numerical data quite well, with a Pearson 335 correlation coefficient R = 0.94 and a root mean square error RMSE = 1.05 days (see 336 Wilks (2011) for the definition of R and RMSE), with the latter representing 7% of the 337 difference between the largest and lowest T_r (15 days). Similar results are obtained for 338 the exposure time (Figure 6b), with R = 0.95 and RMSE = 1.58 days, which repre-339 sents 6% of the difference between the largest and lowest T_e (25 days). These results re-340 flects the capacity of the wind-based model to capture the seasonality, the energy trans-341 fer of most of the anomalous autumn-winter seasons to T_r and T_e , and some of the small 342 spring-summer E fluctuations that modify both time scales during these seasons. 343

An exponential relationship between the residence time and the local forcing was 344 also found in the Pearl River estuary (Sun et al., 2014), but with the freshwater discharge 345 as predictor in this riverine dominated estuary. The exponential model used in their study 346 and in ours captures the asymptotic behaviour of the TTS keeping physical values larger 347 than zero during strong forcing conditions. For our case, the wind-based model can pro-348 vide robust predictions if, for example, the model would be exposed to larger \tilde{E} values 349 not seen during the fitting step. These attributes are hard to achieve with linear or poly-350 nomial models, which make the exponential one a good and a simple tool to predict TTS. 351

An example of the ability of the *wind-based model* to capture anomalous \tilde{T}_r values is the winter of 1990. During this period, the lowest \tilde{T}_r is well reproduced, which is related to the largest \tilde{E} (3 MJ) of our 36-year record (Figure 6a). On the opposite side, we have the winter of 1996, which is a famous period in the North Sea region due to its



Figure 5. As in Figure 4, but for the exposure time.



Figure 6. (a) Time series of the half-year low-pass filter of the spatially-averaged 15-day-mean residence time (\tilde{T}_r) , which is the same as the red line in Figure 5a; the reconstruction of \tilde{T}_r using the wind-based model $(\overline{T}_r, \text{ equation } (4))$; and the sum of the half-year low-pass filter of the 15-day-mean wind energy of the dominant wind sectors W+SW+S (\tilde{E}) . (b) Time series of the half-year low-pass filter of the spatially-averaged 15-day-mean exposure time (\tilde{T}_e) ; the reconstruction of \tilde{T}_e using the wind-based model $(\overline{T}_e, \text{ instead of } \overline{T}_r \text{ in equation } (4))$; and \tilde{E} .

low temperatures (Loewe, 1996). In this season, winds from the most dominant directions were unusually weak, but strong E winds were predominant, with most of their variability contained in periods of less than half a year. During this winter, \tilde{T}_r shows larger values than expected from the climatological winter months and exhibited closer values

to the climatological summer months. The wind-based model (Figures 6a and 6b for \overline{T}_r



Figure 7. Mean sea level pressure and wind at 10 m above ground for a) autumn-winter (September-February) and b) spring-summer (March-August). These averages were obtained using the monthly NCEP/NCAR Reanalysis 1 data for the 1980-2015 period. The mean wind vector is obtained by separately computing the average wind direction and speed following Farrugia and Micallef (2017). The small purple rectangle in panels (a)-(b) represents the DWS numerical domain. The location of the Azores High (AH) and Icelandic Low (IL) pressure systems are also highlighted.

and \overline{T}_{e} , respectively) suggests that the large values of the LTTS in winter of 1996 are explained by the weak wind energy from the usually dominant directions and not by the strong easterly winds observed (which are not explicitly included in the *wind-based model*).

364 365

3.3 The role of the large-scale atmospheric circulation and patterns on the system-wide LTTS

The annual cycle of the large-scale wind in the subtropical North Atlantic is related 366 to the seasonality (a meridional shift and change in intensity) of the the Azores High and 367 the semi-permanent Icelandic Low North Atlantic pressure systems (Trenberth et al., 1990) 368 (Figure 7). This variability is transferred to the regional wind, which induces a local wind 369 response, and ultimately to the LTTS. As a result, a prevailing climatological wind en-370 ergy (from the SW quadrant) is induced in the DWS, which was computed fitting the 371 sum of the 15-day-mean wind energy of the dominant sectors (W+SW+S), see equation 372 (A2) in appendix A for the formal definition) to a model with a free constant and an an-373 nual harmonic. This signal is aligned with the geographical orientation of the system, 374 and characterized by larger values in autumn-winter than in spring-summer (seven times 375 more when contrasting the peaks in November-January with the lowest values in June-376 July, see Figure 3). Thus, it explains why the DWS is at its most efficient climatolog-377 ical state for flushing in autumn-winter, which are the seasons when the LTTS are the 378 lowest (Figures 2a, 2b, and 3). 379

Other low-frequency variations of the wind and sea level pressure, which are not explained by the seasonality, are mainly related to large-scale atmospheric patterns, such as the NAO, EAP and SCAN (Frederikse & Gerkema, 2018). Therefore, our final objective is to determine if interannual variations of the LTTS in the DWS are driven by



Figure 8. The three leading modes of the empirical orthogonal function (EOF) analysis based on the deseasonalized monthly-mean sea level pressure over the North Atlantic sector. (a)-(c) EOFs with units of Pressure (Pa), and (d) the monthly PCs (dotted lines) and their half-year low-pass filtered component (thick lines). The EOF and PC modes are defined following the common positive convention for NAO, EAP and SCAN (see the website of the Climate Prediction Center, https://www.cpc.ncep.noaa.gov/data/teledoc/telecontents.shtml). The first two EOFs (NAO and EAP) are displayed during their positive phases, whereas the third one (SCAN) is depicted during its negative phase. The geostrophic winds computed from the EOFs are depicted with arrows. The variance of the three monthly PCs (PC1 for NAO, PC2 for EAP, and PC3 for SCAN) is scaled to 1, and the numbers in the legend of (d) highlight the fraction of variance explained by the low-pass filtered PCs with respect to their monthly values. The small purple rectangle in panels (a)-(c) represents the DWS numerical domain.

these large-scale patterns. First, we obtain the three leading modes of variability from 384 the EOF analysis of the deseasonalized monthly-mean SLP in the North Atlantic region 385 (see section 2.3.2). Their spatial structure (EOFs) and their temporal variations (PCs) 386 are shown in Figure 8, and they are very similar to those showed by Chafik et al. (2017) 387 and Frederikse and Gerkema (2018). These first three modes at a monthly scale explain 388 32%, 17% and 15% of the SLP variability in the North Atlantic domain. They exhibit 389 large-scale atmospheric structures that are akin to the NAO, EAP and SCAN telecon-390 nection patterns. In comparison to the method used by the Climate Prediction Center 391 (CPC, https://www.cpc.ncep.noaa.gov/data/teledoc/telecontents.shtml), our 392 EOFs and the CPC teleconnection patterns are quite similar, but our PCs and the CPC 393 indices are not necessarily fully interchangeable (Frederikse & Gerkema, 2018). Our first 394 mode (NAO) is characterized by a north-south dipole between the Icelandic low and the 395 Azores high, and it enhances the intensity of the westerlies in the North Sea basin dur-396 ing its positive phase (Figure 8a), whereas the opposite holds during its negative one. 397 Our second mode (EAP) highlights a strong monopole pressure core south of Iceland with 398 meridionally oriented geostrophic winds in the North Sea (Figure 8b). Two weak cores 399 of the opposite sign are also present in the southern part of the subtropical North At-400 lantic region and over Eastern Europe respectively. The NAO and EAP teleconnection 401 patterns modulate the variations in the speed of the jet stream, whereas the NAO mostly 402 describes the latitudinal shifts of the jet, and hence, the main Atlantic storm track (Woollings 403 & Blackburn, 2012). Our third mode (SCAN) displays a zonal pressure dipole between 404 Greenland and Scandinavia, with the strongest center of action over Scandinavia and with 405 a southeastward extension from Greenland towards the Iberian Peninsula (Figure 8c). 406 Its associated geostrophic winds exhibit a strong meridional shear in the North Sea with 407 zonal orientation over much of Western Europe. A positive SCAN is closely related to 408 the well-known Scandinavian blocking weather regime, which in combination with per-409 sistent negative NAO phases, can induce extreme cold outbreaks in Europe during win-410 ter (Cattiaux et al., 2010; Kautz et al., 2020). 411

To link the interannual variations of \tilde{T}_r (and \tilde{T}_e) to the large-scale patterns, we re-412 move the seasonal component from T_r , and then this deseasonalized or anomalous T_r was 413 reconstructed using a multi-linear regression model. The predictors are based on the monthly 414 PCs, which were interpolated to match the 15-day resolution of both LTTS, and then 415 low-pass filtered using a cutoff period of half-year to remove high-frequency variations. 416 We call this reconstruction the *PCs model*. Similar to the *wind-based model*, the PCs of 417 the next 15-day interval are used as predictors. The reconstruction of T_r is obtained by 418 joining the seasonal component with the *PCs model*. This combination is referred to as 419 the *large-scale model* and is shown in Figure 9a; whereas the reconstruction of the de-420 seasonalized T_r given by the *PCs model* is shown in Figure 9b. The *large-scale model* matched 421 T_r quite well, with R = 0.94 and RMSE = 1.03 days. It also explains 96% of the vari-422 ance of T_r (VAR_{exp} in Figure 9c), from which 72% is attributed to the seasonality, 21% 423 to SCAN and NAO, and the remaining 3% to EAP. In general, the model captures most 424 of the autumn-winter variability, but it has difficulties in reproducing the variations of 425 the spring-summer peaks (Figure 9a), as was also the case for the wind-based model (Fig-426 ure 6a). Similar results (R = 0.92 and RMSE = 1.97 days) and weak spring-summer 427 predictability for the *large-scale model* are obtained for T_e (Figure 10). 428

The maximum predictability of the *PCs model* in terms of VAR_{exp} and *R* is found 429 between November and February (Figure 9d), and it is mainly attributed to SCAN and 430 NAO. This behavior is expected since the effects of the large-scale patterns are notice-431 able when the PCs show strong changes and largest values, which is more common dur-432 ing autumn-winter (Figure 8d). The lowest T_r observed during autumn-winter (Figure 433 9a) are predominantly associated with the interplay between negative SCAN, positive 434 NAO, and positive EAP (Figure 8d), with the latter having the lowest contribution. The 435 combination of their spatial patterns tend to induce along-coast anomalous winds (mostly 436 from W and SW directions) that favor the flushing efficiency of the DWS system. This 437



Figure 9. (a) Time series of the half-year low-pass filter of the spatially-averaged 15-day-mean residence time (\tilde{T}_r) , its seasonal component, and its reconstruction with the *large-scale model* (seasonal + *PCs model*). (b) The deseasonalized \tilde{T}_r and its reconstruction with the *PCs model*. (c) Explained variance VAR_{exp} and correlation *R* for the reconstruction of \tilde{T}_r using cumulative components of the *large-scale model*: only the seasonal component; seasonal + SCAN (+SCAN); seasonal + SCAN + NAO (+NAO); and the *large-scale model*, i.e., seasonal + SCAN + NAO + EAP (+EAP). The VAR_{exp} is defined as the ratio between the variance of the cumulative components of the *large-scale model* and the variance of \tilde{T}_r . (d) Monthly statistics (VAR_{exp} and *R*) of the reconstruction of the deseasonalized \tilde{T}_r with the *PCs model*. In this case, the VAR_{exp} is defined as the ratio between the variance of the deseasonalized \tilde{T}_r per month.

behavior is consistent with the study of Chafik et al. (2017), in which negative SCAN and positive NAO patterns explain most anomalous high monthly sea level values observed at several North Sea tidal gauge stations during autumn-winter. According to our results, they are concurrent with strong flushing conditions and with a low likelihood for the particles to return to the DWS (represented by low \tilde{T}_r and \tilde{T}_e , respectively).

Winters with the strongest flushing were well captured by the *PCs model* (Figures 443 9a and 9b). For example, in the winters of 1990, 1995, 2007, and 2014, a decrease of T_r 444 of around 3-5 days with respect to the December-January climatological value of 10 days 445 was observed, which was related to high \dot{E} (Figure 6a). Therefore, the lowest T_r values 446 were induced by large-scale atmospheric patterns and not by storms, which induce high-447 frequency variations and are commonly associated with the presence of well-known weather 448 regimes (Hochman et al., 2021). For example, during the well-known winter of 1990, two 449 exceptionally strong storms ("Daria" and "Vivian") passed over central Europe and crossed 450 the North Sea in just few days (Pinto et al., 2009). As a result, they trigger the strongest 451 hourly wind speeds from SW and W directions in our 36-year record (around 30 m/s or 452 60 MJ), but induced 15-day-mean peaks in the wind energy similar to other less stormy 453 periods. On the other hand, the most anomalous winters with the largest T_r values (1996) 454 and 2006) were also well explained by the PCs model. During these winters, an increase 455



Figure 10. As in Figure 9, but for the exposure time.

of T_r of about 7 and 4 days with respect to the December-January climatology were ob-456 served. However, the PCs model underestimates these values by around 2-3 days and 457 1 day, respectively. The most extreme change between two consecutive winters (10 days 458 for \tilde{T}_r and about 20 days for \tilde{T}_e) occurred between the winters of 1995-1996. In 1995, 459 a combination of negative SCAN with positive NAO and EAP triggered a \tilde{E} stronger 460 than its December-January climatology and induced one of the lowest T_r (around 6 days). 461 The following year, the largest \tilde{T}_r during winter was observed (about 17 days). During 462 this famous winter, positive SCAN and negative NAO induced strong E winds. However, 463 as was stated in the previous section, the lack of E (and hence, the background forcing 464 by the tides and freshwater discharge) is enough to explain why T_r during the 1996 win-465 ter was similar to its May-July climatology. In agreement with this, during the winter 466 of 2006 (and to a lesser extent for 2003, 2009, and 2010), T_r was also larger than its cli-467 matological value, which was related to quite low \tilde{E} , but also to low-frequency energy 468 from the other directions. 469

470

3.4 Other sources of variability on the LTTS

Variations of the bathymetry were neglected in our simulation, which was done in-471 tentionally to isolate the role of the atmospheric forcing in our results. Relative stabil-472 ity in the location and orientation of the major channels connected to the Texel inlet has 473 been observed since approximately 1972 or 40 years after the construction of the Afs-474 luitdijk in 1932 (Elias et al., 2003, 2006), which is a closure dyke of around 30 km where 475 the two main sluices feeding freshwater into the are located (Figure 1). Changes in the 476 sedimentation-erosion patterns of the channels were observed in these studies, but with 477 only minor modifications of the bathymetry profiles. Thus, during our period of anal-478 ysis, we expect small effects of these bathymetry variations compared to the large effects 479 of wind, particularly when focusing on the variability of system-wide LTTS, as is the case 480 in most of our results. 481

The time series of the freshwater discharge from the sluice located at Den Oever is correlated with \tilde{E} (R = 0.56) and anti-correlated with \tilde{T}_r (R = -0.68) and \tilde{T}_e (R =

-0.61); whereas for the sluice located at Kornwerderzand non-significant correlations are 484 obtained. Because of this, it is not trivial to isolate the effect of both sluices on the vari-485 ability of T_r and T_e under our current approach. However, it is known that the resid-486 ual volume flow rate through the DWS during strong wind conditions from the domi-487 nant directions is one order of magnitude larger than the one associated with the tides 488 and the freshwater discharge (Duran-Matute et al., 2014), and that the total freshwa-489 ter discharge of both sluices can only explain less than 5% of the variability of the resid-490 ual transport in this system (Donatelli et al., 2022a). Therefore, we expect that the fresh-491 water discharge and the residual tidal currents are the main factors controlling the back-492 ground T_r (A = 19.29 days) and T_e (A = 27.96 days), which are obtained when the 493 wind energy of the most energetic sectors (\vec{E}) is null in the *wind-based model*. In addi-101 tion, these forcing mechanisms also seem to explain part of the variability of T_r and T_e 495 not explained by the *wind-based model* and the *large-scale model* during calm conditions 496 (mainly spring-summer months), which are the periods in which both these models show 497 strong lack of predictability. 498

Because our main results are based on the characterization of the system-wide LTTS, 499 the vertical structure of the LTTS was ignored using depth-averaged currents. Locally, 500 there might be a marked heterogeneity in this vertical structure (Wolk, 2003; Du & Shen, 501 2016), which might be associated to, for example, a strong gravitational circulation. How-502 ever, it is not currently feasible to perform a 3D Lagrangian analysis for 36-year of the 503 DWS due to the amount of data required to compute the necessary 3D particle trajec-504 tories. Nonetheless, our results can be useful to select, simulate, and understand the 3D-505 behaviour of the LTTS during particular and striking conditions, like the transition be-506 tween the winters with strong and weak winds from the most energetic directions in 1995-507 1996. 508

509 4 Conclusions

While it has been acknowledged that high-frequency events, like storms crossing 510 the Dutch Wadden Sea (DWS) in few days or bora winds in the Venice lagoon, can com-511 pletely renew the water in multiple-inlet systems, we show here that low-frequency wind 512 variability can also play a large role in modulating the transport time scales in a multiple-513 inlet system. The broad and immediate implication of our results is that interannual changes 514 in the atmospheric patterns can have a much larger effect on the variations of the wa-515 ter transport than may have been expected, and hence, on the long-term ecology and 516 functioning of multiple-inlet systems. 517

For the case of the DWS, the lowest system-wide Lagrangian transport time scales 518 (LTTS) are observed in several years during autumn-winter months and are well explained 519 by the concurrent negative phase of the Scandinavia Pattern (SCAN) and the positive 520 phase of the North Atlantic Oscillation (NAO), which induce stronger SW and W winds 521 in this system. These winds trigger an anomalous eastward flow that enhances the flush-522 ing efficiency, which is typically already strong in autumn-winter. The opposite happens 523 during positive SCAN and negative NAO, and weaker flushing during autumn-winter 524 is observed. In contrast to single-inlet systems (like in the study of Du and Shen (2016)), 525 our results show that system-wide LTTS in multiple-inlet systems, like the DWS, are rep-526 resentative of the overall system when studying the influence of winds on the seasonal 527 and interannual variations of the LTTS. This response is in agreement with the fact that 528 winds from specific intensities and directions are very efficient in forcing net residual trans-529 port across watersheds (i.e. tidal divides) and trough the inlets of multiple-inlet systems 530 (Li, 2013; Duran-Matute et al., 2016). A similar response can be expected in other wind-531 dominated multiple-inlet systems (e.g., along the North Sea coast), leading to seasonal 532 and interannual variations of the LTTS driven by the large-scale circulation and atmo-533 spheric patterns, respectively. 534

Our findings also reveal that care should be taken when observing variations of the 535 long-term values of the residual volume flow rate across inlets and watersheds, when events 536 with strong wind conditions from the analysis are removed. Using this approach, Donatelli 537 et al. (2022a) found changes of the long-term residual volume transport using a 11-year 538 simulation of the DWS. According to our results, this does no necessarily indicate that 539 extreme events can alter these long-term values. Instead, we expect that long-term vari-540 ations of the residual flow rate in other wind-dominated multiple-inlet systems would be 541 also driven by large-scale atmospheric patterns, as was the case for the interannual vari-542 ations of the LTTS in the DWS (from our current study), and for the multi-decadal sea 543 level variability along the North Sea coastal areas (Frederikse & Gerkema, 2018). 544

Finally, our study highlights the importance of understanding the water transport 545 variability due to local and remote forcing to, for example, explain better why large-scale 546 atmospheric patterns affect biological processes (see e.g., Straile & Adrian, 2000; Gol-547 ubkov & Golubkov, 2021), and to improve analytical models that use TTS to model eco-548 logical processes (see e.g., Lucas & Deleersnijder, 2020). From a practical point of view, 549 analytical models like those proposed here to predict the LTTS using the wind and the 550 large-scale atmospheric patterns could be employed to estimate the LTTS during peri-551 ods not covered by such detailed simulations, particularly, for seasonal forecasts and fu-552 ture climate-change scenarios. 553

554 Conflict of Interest

555

The authors declare no conflicts of interest relevant to this study.

556 Data Availability Statement

Data and scripts (based on Python v3.8) used to reproduce the figures of this study 557 are available at the GitHub repository https://github.com/JeancarloFU/paper_Atmospherically 558 _Driven_Seasonal_Interannual_LTTS_MultipleInlet. The wavelet analysis is based 559 on the Python package Pycwt v0.3.0a22 (https://anaconda.org/conda-forge/pycwt), 560 but we added a script to perform the bias correction (Liu et al., 2007) and a wavelet fil-561 ter (Torrence & Compo, 1998). Monthly-mean sea level pressure and wind at 10 m above 562 ground were obtained from The NCEP-NCAR Reanalysis 1 data, which is provided by 563 the NOAA PSL, Boulder, Colorado, USA, from their website at https://psl.noaa.gov/ 564 data/gridded/data.ncep.reanalysis.html. Eulerian data was produced with the GETM/GOTM model, and its set-up is described in Duran-Matute et al. (2014) and Gräwe et al. (2016). 566 The Lagrangian model (Parcels v2.1.1) can be downloaded from https://anaconda.org/ 567 conda-forge/parcels or https://oceanparcels.org. 568

569 Acknowledgement

This study was financially supported by the NWO/ENW project: 'The Dutch Wadden Sea as an event-driven system: long-term consequences for exchange (LOCO-EX)'

- (OCENW.KLEIN.138). The numerical simulations with GETM were done thanks to the
- ⁵⁷³ North-German Supercomputing Alliance (HLRN).

574 Appendix A Wind energy averaging

To establish connections with the LTTS and to smooth the noisy, hourly, high-resolution wind energy data from equation (3), and to remove most of the high-frequency effects (e.g., storms), we compute the mean wind energy during 15-day intervals. For a given

sectorial direction s, the 15-day-mean wind energy is defined as

$$E_s = N^{-1} \sum_n E_{s,n} = \frac{1}{2} \rho A \Delta t N^{-1} \sum_n W_{s,n}^3 = C \Delta t N^{-1} \sum_n W_{s,n}^3, \tag{A1}$$

where N = 360 is the total amount of hourly data points, $C = \frac{1}{2}\rho A = 0.6125$ kg m⁻¹, and the total wind energy is obtained from $E_T = \sum_s E_s = C\Delta t N^{-1} \sum_n W_n^3$, where $W_n^3 = \sum_s W_{s,n}^3$ is the cube of the hourly wind speed. A similar expression to equation (A1), but for yearly averages, was used by Gerkema and Duran-Matute (2017) and Donatelli et al. (2022a).

The sum of the wind energy of the most energetic sectors (W+SW+S) is obtained from equation (A1) yielding

$$E = E_W + E_{SW} + E_S. \tag{A2}$$

This time series, with 15-day resolution, was employed to compute the annual cycle showed in Figure 3.

Then, we apply a half-year low-pass filter to each E_s (equation (A1)), as was done 582 for the LTTS, which we call E_s . Due to the undulatory nature of the wavelet filter (and 583 other similar ones like the Lanczos filter) and to the fact that E_s could be near zero, slightly 584 negative values appear. To be physically correct, we set all negative values of E_s to zero. 585 Finally, we add E_s from the most energetic sectors (W+SW+S), and get E, which we 586 call the sum of the half-year low-pass filter of the 15-day-mean wind energy of the dom-587 inant sectors. Almost identical results are obtained if we apply the low-pass filter directly 588 to E defined in equation (A2). 589

590 References

- ⁵⁹¹ Burchard, H., & Bolding, K. (2002). GETM a general estuarine transport model. ⁵⁹² Scientific documentation, Tech. Rep. EUR, 20253 EN.
- Cattiaux, J., Vautard, R., Cassou, C., Yiou, P., Masson-Delmotte, V., & Codron,
 F. (2010). Winter 2010 in europe: A cold extreme in a warming climate.
 Geophysical Research Letters, 37(20).
- Chafik, L., Nilsen, J. E. Ø., & Dangendorf, S. (2017). Impact of North Atlantic tele connection patterns on Northern European sea level. Journal of Marine Sci ence and Engineering, 5(3), 43. doi: https://doi.org/10.3390/jmse5030043
- Cimatoribus, A., Lemmin, U., & Barry, D. (2019). Tracking Lagrangian transport in
 Lake Geneva: A 3D numerical modeling investigation. *Limnology and Oceanog- raphy*, 64(3), 1252–1269. doi: https://doi.org/10.1002/lno.11111
- Cucco, A., & Umgiesser, G. (2006). Modeling the Venice Lagoon residence time. *Ecological modelling*, 193(1-2), 34–51. doi: https://doi.org/10.1016/j.ecolmodel .2005.07.043
- ⁶⁰⁵ Cucco, A., Umgiesser, G., Ferrarin, C., Perilli, A., Canu, D. M., & Solidoro, C.
- 606(2009). Eulerian and lagrangian transport time scales of a tidal active coastal607basin. Ecological Modelling, 220(7), 913–922. doi: https://doi.org/10.1016/608j.ecolmodel.2009.01.008
- Defne, Z., & Ganju, N. K. (2015). Quantifying the residence time and flushing
 characteristics of a shallow, back-barrier estuary: Application of hydrodynamic
 and particle tracking models. *Estuaries and Coasts*, 38(5), 1719–1734. doi:
 https://doi.org/10.1007/s12237-014-9885-3
- Delandmeter, P., & Van Sebille, E. (2019). The Parcels v2. 0 Lagrangian framework:
 new field interpolation schemes. *Geoscientific Model Development*, 12(8),
 3571–3584. doi: https://doi.org/10.5194/gmd-12-3571-2019
- den Heyer, C., & Kalff, J. (1998). Organic matter mineralization rates in sediments:
 A within-and among-lake study. *Limnology and oceanography*, 43(4), 695–705.
 doi: https://doi.org/10.4319/lo.1998.43.4.0695

619	Dippner, J. W., Bartl, I., Chrysagi, E., Holtermann, P., Kremp, A., Thoms, F., &
620	Voss, M. (2019). Lagrangian residence time in the Bay of Gdańsk, Baltic
621	Sea. Frontiers in Marine Science, 725. doi: https://doi.org/10.3389/
622	fmars.2019.00725
623	Donatelli, C., Duran-Matute, M., Gräwe, U., & Gerkema, T. (2022a). Residual
624	circulation and freshwater retention within an event-driven system of inter-
625	tidal basins. Journal of Sea Research, 102242. doi: https://doi.org/10.1016/
626	j.seares.2022.102242
627	Donatelli, C., Duran-Matute, M., Gräwe, U., & Gerkema, T. (2022b). Statistical
628	detection of spatio-temporal patterns in the salinity feld within an inter-
629	tidal basin. Estuaries and Coasts, 1–18. doi: https://doi.org/10.1007/
630	s12237-022-01089-3
631	Du, J., & Shen, J. (2016). Water residence time in Chesapeake Bay for 1980–2012.
632	Journal of Marine Systems, 164, 101–111. doi: https://doi.org/10.1016/
633	j.jmarsys.2016.08.011
634	Duran-Matute, M., Gerkema, T., De Boer, G., Nauw, J., & Gräwe, U. (2014).
635	Residual circulation and freshwater transport in the Dutch Wadden Sea:
636	a numerical modelling study. $Ocean Science, 10(4), 611-632.$ doi:
637	https://doi.org/10.5194/os-10-611-2014
638	Duran-Matute, M., Gerkema, T., & Sassi, M. G. (2016). Quantifying the residual
639	volume transport through a multiple-inlet system in response to wind forcing:
640	the case of the western Dutch Wadden Sea. <i>Journal of Geophysical Research:</i>
641	Oceans, 121(12), 8888–8903, doi: https://doi.org/10.1002/2016JC011807
642	Elias, E., Cleveringa, J., Buijsman, M., Roelvink, J., & Stive, M. (2006). Field and
643	model data analysis of sand transport patterns in texel tidal inlet (the nether-
644	lands). Coastal Engineering, 53(5-6), 505–529. doi: https://doi.org/10.1016/
645	j.coastaleng.2005.11.006
646	Elias, E., Stive, M., Bonekamp, H., & Cleveringa, J. (2003). Tidal inlet dynamics in
647	response to human intervention. Coastal engineering journal, 45(04), 629–658.
648	doj: https://doi.org/10.1142/S0578563403000932
649	Farrugia, P. S., & Micallef, A. (2017). Vectorial statistics for the standard deviation
650	of wind direction. Meteorology and Atmospheric Physics, 129(5), 495–506, doi:
651	https://doi.org/10.1007/s00703-016-0483-8
652	Frederikse T & Gerkema T (2018) Multi-decadal variability in seasonal mean
653	sea level along the North Sea coast. Ocean Science, 14(6), 1491–1501. doi:
654	https://doi.org/10.5194/os-14-1491-2018
655	Gerkema T (2019) An introduction to tides Cambridge University Press
656	Gerkema, T. & Duran-Matute M (2017) Interannual variability of mean sea
657	level and its sensitivity to wind climate in an inter-tidal hasin <i>Earth System</i>
658	Dunamics = 8(4) = 1223 - 1235 doi: https://doi.org/10.5194/esd-8-1223-2017
650	Colubkov M & Colubkov S (2021) Relationships between northern hemi-
660	sphere teleconnection patterns and phytoplankton productivity in the Neva
661	Estuary (northeastern Baltic Sea) Frontiers in Marine Science 1339 doi:
662	https://doi.org/10.3389/fmars.2021.735790
662	Conzález F II T Herrera-Silveira I A & Aguirre-Macedo M I. (2008) Water
664	quality variability and eutrophic trends in karstic tropical coastal lagoons of
665	the Vucatán Peninsula Estuarine Coastal and Shelf Science $76(2)$ 418–430
666	doi: https://doi.org/10.1016/j.ecss.2007.07.025
667	Cräwe II Flöser C. Cerkema T. Duran-Matute M. Badewien T. H. Schulz F.
007	& Burchard H (2016) A numerical model for the entire Wedden Soe: Skill
000	assessment and analysis of hydrodynamics Journal of Coonhusical Research.
670	O_{ceans} 121(7) 5231–5251 doi: https://doi.org/10.1002/2016.JC011655
671	Hajlegeorgis D. Lachkar Z. Riener C. & Gruber N. (2021) \triangle Lagrangian
672	study of the contribution of the Canary coastal unwelling to the nitrogen
672	budget of the open North Atlantic $Biogeosciences 18(1) 303-325$ doi:
015	\mathcal{D}

674	https://doi.org/10.5194/bg-18-303-2021
675	Herrling, G., & Winter, C. (2015). Tidally-and wind-driven residual circulation at
676	the multiple-inlet system east frisian wadden sea. Continental Shelf Research,
677	106, 45–59. doi: https://doi.org/10.1016/j.csr.2015.06.001
678	Hochman, A., Messori, G., Quinting, J. F., Pinto, J. G., & Grams, C. M. (2021). Do
679	Atlantic-European weather regimes physically exist? Geophysical Research Let-
680	ters, 48, e2021GL095574. doi: https://doi.org/10.1029/2021GL095574
681	Huguet, JR., Brenon, I., & Coulombier, T. (2019). Characterisation of the wa-
682	ter renewal in a macro-tidal marina using several transport timescales. Water,
683	11(10), 2050. doi: https://doi.org/10.3390/w11102050
684	Jiang, L., Soetaert, K., & Gerkema, T. (2019). Decomposing the intra-annual vari-
685	ability of flushing characteristics in a tidal bay along the North Sea. Journal of
686	Sea Research, 155, 101821. doi: https://doi.org/10.1016/j.seares.2019.101821
687	Kalnay, E., Kanamitsu, M., Kistler, R., Collins, W., Deaven, D., Gandin, L.,
688	others (1996). The NCEP/NCAR 40-year reanalysis project. Bulletin of the
689	American meteorological Society, 77(3), 437–472. doi: https://doi.org/10.1175/
690	1520-0477(1996)077(0437:TNYRP)2.0.CO:2
691	Kautz, LA., Polichtchouk, L., Birner, T., Garny, H., & Pinto, J. G. (2020). En-
692	hanced extended-range predictability of the 2018 late-winter eurasian cold spell
693	due to the stratosphere. Quarterly Journal of the Royal Meteorological Society.
694	146(727), 1040–1055, doi: https://doi.org/10.1002/gi.3724
695	Lange, M., & van Sebille, E. (2017). Parcels v0.9: prototyping a Lagrangian ocean
696	analysis framework for the petascale age. Geoscientific Model Development.
697	10(11), 4175–4186, doi: https://doi.org/10.5194/gmd-10-4175-2017
698	Li, C. (2013). Subtidal water flux through a multiple-inlet system: Observations be-
699	fore and during a cold front event and numerical experiments. <i>Journal of Geo</i>
700	<i>physical Research: Oceans. 118</i> (4), 1877–1892. doi: https://doi.org/10.1002/
701	igrc.20149
702	Liu, Y., San Liang, X., & Weisberg, R. H. (2007). Rectification of the bias in the
703	wavelet power spectrum. Journal of Atmospheric and Oceanic Technology.
704	24(12), 2093–2102. doi: https://doi.org/10.1175/2007JTECHO511.1
705	Loewe, P. (1996). Surface temperatures of the North Sea in 1996. Deutsche
706	Hudrografische Zeitschrift, 48(2), 175–184. doi: https://doi.org/10.1007/
707	BF02799386
708	Lucas, L. V., & Deleersniider, E. (2020). Timescale methods for simplify-
709	ing, understanding and modeling biophysical and water quality processes
710	in coastal aquatic ecosystems: a review. $Water, 12(10), 2717.$ doi:
711	https://doi.org/10.3390/w12102717
712	Monsen, N. E., Cloern, J. E., Lucas, L. V., & Monismith, S. G. (2002). A
713	comment on the use of flushing time, residence time, and age as trans-
714	port time scales. Limnology and oceanography, 47(5), 1545–1553. doi:
715	https://doi.org/10.4319/lo.2002.47.5.1545
716	Patgaonkar, R. S., Vethamony, P., Lokesh, K., & Babu, M. (2012). Residence time
717	of pollutants discharged in the Gulf of Kachchh, northwestern Arabian Sea.
718	Marine pollution bulletin, 64(8), 1659–1666. doi: https://doi.org/10.1016/
719	i.marpolbul.2012.05.033
720	Pawlowicz, R., Hannah, C., & Rosenberger, A. (2019). Lagrangian observations
721	of estuarine residence times, dispersion, and trapping in the Salish Sea. Estu-
722	arine. Coastal and Shelf Science, 225, 106246. doi: https://doi.org/10.1016/j
723	.ecss.2019.106246
724	Pinto, J. G., Zacharias, S., Fink, A. H., Leckebusch, G. C. & Ulbrich U (2009)
725	Factors contributing to the development of extreme North Atlantic cyclones
726	and their relationship with the NAO. <i>Climate dynamics</i> , 32(5), 711–737. doi:
727	https://doi.org/10.1007/s00382-008-0396-4
728	Rayson, M. D., Gross, E. S., Hetland, R. D., & Fringer, O. B. (2016). Time scales

729	in Galveston Bay: An unsteady estuary. Journal of Geophysical Research:
730	Oceans, 121(4), 2268–2285. doi: https://doi.org/10.1002/2015JC011181
731	Ridal, M., Olsson, E., Unden, P., Zimmermann, K., & Ohlsson, A. (2017). Un-
732	certainties in ensembles of regional re-analyses. Deliverable D2.7 HAR-
733	MONIE Reanalysis Report of Results and Dataset. Retrieved 2022-01-15,
734	from http://www.uerra.eu/component/dpattachments/?task=attachment
735	.download&id=296
736	Ridderinkhof, H., & Zimmerman, J. (1990). Mixing processes in a numerical model
737	of the Western Dutch Wadden Sea. In Residual currents and long-term trans-
738	<i>port</i> (p. 194-209). Springer. doi: https://doi.org/10.1007/978-1-4613-9061-9
739	15
740	Safak, I., Wiberg, P., Richardson, D., & Kurum, M. (2015). Controls on residence
741	time and exchange in a system of shallow coastal bays. Continental Shelf Re-
742	search. 97. 7–20. doi: https://doi.org/10.1016/i.csr.2015.01.009
743	Schwichtenberg, F., Callies, U., & van Beusekom, J. E. (2017). Residence times
744	in shallow waters help explain regional differences in Wadden Sea eutrophi-
745	cation Geo-Marine Letters 37(2) 171–177 doi: https://doi.org/10.1007/
746	s00367-016-0482-2
740	Soomere T Andreiev O Myrberg K & Sokolov A (2011) The use of La-
749	grangian trajectories for the identification of the environmentally safe fairways
740	grangian trajectories for the identification of the environmentally sale ran ways. Marine Pollution Bulletin $62(7)$ $1/10-1/20$ doi: https://doi.org/10.1016/
749	i marnelbul 2011 0.4 0.41
750	Straile D & Adrian B (2000) The North Atlantic Oscillation and plank-
751	ton dynamics in two European lakes: two variations on a general theme
752	Clobal Change Biology 6(6) 663–670 doi: https://doi.org/10.1046/
753	$i 1365_{-2}486 2000 00350 \text{ v}$
754	Sun I Lin B Li K k ligng C (2014) A modelling study of residence time and
755	ovposure time in the poorl river estuary china Lowral of Hudro environment
750	Research 8(3) 281–201 doi: https://doi.org/10.1016/j.jbor.2013.06.003
757	Torrence C is Compo C P (1008). A practical guide to wavelet analysis Bulletin
758	Torrence, C., & Compo, G. I. (1996). A practical guide to wavelet analysis. Duttettil of the American Mateorelogical accients $70(1)$ 61–78 doi: https://doi.org/10
759	1175/1520 0.077(1008)070/0061 A DCTWA 2.0 CO 2
760	Tranbarth K F Large W C ℓ Olson I C (1900) The mean annual cycle
701	in global ocean wind stress Journal of Physical Ocean or annual cycle
762	1760 doi: https://doi.org/10.1175/1520.0485(1000)020/1742:TMACIC\2.0.CO.
763	$\frac{1100. \text{ doi. https://doi.org/10.1113/1520-0465(1990)020(1142.1101AC1G/2.0.00)}{2}$
764	Willis D S (2011) Statistical methods in the atmospheric sciences (Vol. 100) Aco
765	domia pross
766	Woll: E (2003) Three dimensional lagrangian tracer modelling in wadden eeg areas
767	(Diploma thesis) Carl von Ossietzky University Oldenburg
768	Woollings T & Rhelchurn M (2012) The north startig int stream under alignet
769	woonings, 1., & Diackburn, M. (2012). The north attained jet stream under chimate $\theta_{2}^{(2)}$
770	change and its relation to the had and ea patterns. <i>Journal of Cumule</i> , 25(5), 886,002, doi: https://doi.org/10.1175/JCLI.D.11.00087.1
771	$V_{iong} = V_{iong} = 0.11110$ (2021) Weter exchange and its relation
772	tionshing with external foreings and residence time in Checapache Pay
773	Learnal of Marine Sectors 015, 102407
774	<i>Journal of Marine Systems</i> , 215, 105497. doi: https://doi.org/10.1010/
775	J.Jinarsys.2020.105497 Zhang W. C. Willin I. L. & Schafold O. M. (2010). Simulation of motor and
776	Zhang, W. G., Wilkin, J. L., & Schöneid, O. M. (2010). Simulation of water age and
777	residence time in New York Bignt. Journal of Physical Oceanography, $4U(5)$, 065, 082, doi: https://doi.org/10.1175/2000.IDC/4240.1
778	900-982. doi: nttps://doi.org/10.11/5/2009JPO4249.1
779	Zimmerman, J. I. F. (1970). Mixing and flushing of tidal embayments in the
780	we stern Dutch wadden bea part I: Distribution of salinity and calculation of minimating applea. Notherlanda Leave $l = f(C_{12}, P_{12}, \dots, P_{12})$
781	mixing time scales. <i>Netherianas Journal of Sea Kesearch</i> , $IU(2)$, 149–191. doi: https://doi.org/10.1016/0077.7770(76)00012.0
782	nttps://doi.org/10.1016/0077-7579(76)90013-2

Supporting Information for "Atmospherically Driven Seasonal and Interannual Variability in the Lagrangian Transport Time Scales of a Multiple-inlet Coastal System"

Jeancarlo M. Fajardo-Urbina¹, Gijs Arts¹, Ulf Gräwe², Herman J.H. Clercx¹, Theo Gerkema³, Matias Duran-Matute¹

¹Fluid and Flows group and J.M. Burgers Center for Fluid Dynamics, Department of Applied Physics and 7 Science Education, Eindhoven University of Technology, Eindhoven, the Netherlands 8 ²Leibniz Institute for Baltic Sea Research Warnemünde, Rostock, Germany ³Department of Estuarine and Delta Systems, NIOZ Royal Netherlands Institute for Sea Research, 10 Yerseke, the Netherlands

Contents of this file 12

Text S1

Introduction 14

1

2

3

4

5

6

9

11

13

This supporting information contains the text S1 that provide details about the 15 validation of the Eulerian numerical model, which was employed to feed a Lagrangian 16 model to obtain particles trajectories. The references used in Texts S1 are also provided. 17

Text S1 18

Model validation using sea-surface height (SSH), currents, temperature, and salin-19 ity during the year 2009-2010 was done by Duran-Matute et al. (2014) and Gräwe et al. 20 (2016) using similar model configurations. We have validated our simulation only for the 21 years 2009-2010, as was done by Duran-Matute et al. (2014), because the bathymetry 22 is mostly based on those years. Thus, it is expected that the simulation during this time-23 span is most compatible with observations. We contrast our numerical results with SSH 24 measured at 14 tidal stations located within and around the DWS (Figure 1 from the 25 main manuscript), and with the amplitude and phase of the M2, S2 and M4 tides. In 26 general, our simulation shows similar performance to that of Duran-Matute et al. (2014) 27 (see Table 1), which is remarkable because Duran-Matute et al. (2014) used results from 28 a two-dimensional model with data assimilation to impose SSH at the boundaries. The 29 good performance of the model is also reflected in the form factor F=(K1+O1)/(M2+S2), 30 which is defined as the ratio of the amplitudes of the K1 and O1 harmonics to those of 31 the M2 and S2 ones (Defant, 1961). The observations and the simulation by Duran-Matute 32 et al. (2014) yield F=0.166, whereas F=0.175 for our case. We have not validated directly 33 our simulated particles trajectories with observational data in the DWS due to the dif-34 ficulty of acquiring it in shallow coastal systems containing large areas of intertidal flats. 35 However, our model configuration was employed recently by Donatelli et al. (2022a, 2022b), 36 whose results showed good agreement with those of previous numerical setups in the DWS 37 region (Duran-Matute et al., 2014; Gräwe et al., 2016). Therefore, we expect that this 38 performance is also reflected in our Lagragian analysis. 39

Corresponding author: Matias Duran-Matute, m.duran.matute@tue.nl

Table 1. Summary of the Model Skill Assessments of the SSH, and M2, S2 and M4 Tides for the Period $2009-2010^a$

	SSH		M2	S2	M4
\mathbf{R}	0.98(0.99)	\mathbf{MAE}_{amp} (%)	4.3 (6.1)	10.6(4.3)	15.0(8.7)
\mathbf{RMSE} (m)	$0.15\ (0.10)$	\mathbf{MAE}_{pha} (min)	4.6(8.7)	9.6(9.3)	46.5(27.5)
NRMSE $(\%)$	3.5(2.3)				

 a The values displayed in this table represent the average of the 14 tidal stations located around the DWS

(Figure 1 from the main manuscript). In the second column the performance of the SSH is shown in terms of the correlation R, the root mean square error RMSE, and the NRMSE, which is the RMSE normalized with the difference between the largest and lowest observed SSH. In the last three columns the results for the M2, S2, M4 in terms of the mean absolute error MAE are displayed (see Wilks (2011) for the definition of R, RMSE and MAE). The MAE is in percent for the amplitude and in minutes for the phase. In parenthesis, we give the statistics obtained with the simulation of Duran-Matute et al. (2014).

40 References

41	Defant, A. (1961). Physical oceanography (Vol. 1). Pergamon.
42	Donatelli, C., Duran-Matute, M., Gräwe, U., & Gerkema, T. (2022a). Residual
43	circulation and freshwater retention within an event-driven system of inter-
44	tidal basins. Journal of Sea Research, 102242. doi: https://doi.org/10.1016/
45	j.seares.2022.102242
46	Donatelli, C., Duran-Matute, M., Gräwe, U., & Gerkema, T. (2022b). Statistical
47	detection of spatio-temporal patterns in the salinity feld within an inter-
48	tidal basin. Estuaries and Coasts, 1–18. doi: https://doi.org/10.1007/
49	s12237-022-01089-3
50	Duran-Matute, M., Gerkema, T., De Boer, G., Nauw, J., & Gräwe, U. (2014).
51	Residual circulation and freshwater transport in the dutch wadden sea:
52	a numerical modelling study. $Ocean Science, 10(4), 611-632.$ doi:
53	https://doi.org/10.5194/os-10-611-2014
54	Gräwe, U., Flöser, G., Gerkema, T., Duran-Matute, M., Badewien, T. H., Schulz, E.,
55	& Burchard, H. (2016). A numerical model for the entire wadden sea: Skill
56	assessment and analysis of hydrodynamics. Journal of Geophysical Research:
57	Oceans, 121(7), 5231–5251. doi: https://doi.org/10.1002/2016JC011655
58	Wilks, D. S. (2011). Statistical methods in the atmospheric sciences (Vol. 100). Aca-
59	demic press.